PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

Title (Provisional)

PROTOCOL FOR A RANDOMIZED CONTROLLED TRIAL OF A WEIGHT MAINTENANCE INTERVENTION TO PROMOTE FAT LOSS IN PREGNANT INDIVIDUALS WITH OBESITY.

Authors

Redman, Leanne; Phelan, Suzanne; Apolzan, John; Beyl, Robbie; Altazan, Abby; Dickey, Madison; Simeon, Emerson; Flanagan, Emily W.; Cabre, Hannah E.; Sparks, Joshua R.; Kebbe, Maryam; Caughey, AB; Valent, Amy M.; Hsia, Daniel S.; Yin, Elaine; Keadle, Sarah

This article was not externally reviewed at BMJ Open. Protocols that have been independently assessed prior to submission to BMJ Open are usually fast-tracked to publication on the grounds that further substantial changes will not be possible. This independent assessment will usually be external, independent review for both a substantial grant award from a non-commercial or government funder and ethics approval.